Author's response to reviews

Title: A Phase II Clinical Trial to assess the safety of Clonidine in Acute Organophosphate Poisoning

Authors:

Polwattage M.S Perera (mark@sactrc.org)
Shaluka F Jayammani (shalukajaya@yahoo.com)
Raja Hettiarachchi (raga@sactrc.org)
Nick Buckley (n.buckley@unsw.edu.au)
Chandana Abeysinghe (chandana@sactrc.org)
Harendra Karunathilaka (harendra@sactrc.org)
Andrew H. Dawson (adawson@sactrc.org)

Version: 3 Date: 18 November 2008

Author's response to reviews:

I have attached herewith a phase II clinical Trial entiteld: A Phase II Clinical Trial to assess the safety of Clonidine in Acute Organophosphate Poisoning. With the necessary correction which were:

1. Include a statement of Competing Interests in the manuscript. Declaration of competing interests:

   All authors have no competing interests as defined by Trials Publishing Group, or other interests that might be perceived to influence the results and/or discussion reported in this paper.

2. Please include a section for Abbreviations, if abbreviations are not defined after first use in the manuscript. –Abbreviations were defined see below

   GABA has been removed and replaced with Gamma-aminobutyric acid

   in the aknowledgement section SACTRC has been defined as South Asian Toxicologic Research Collaboration.